Physics 410: Computational Physics (Fall 2004)

Instructor Office Office Hours Office Phone Home Phone : 222-9424 E-mail choptuik@physics.ubc.ca Grader: Bruno Mundim bruno@physics.ubc.ca Course Home Page http://laplace.physics.ubc.ca/410/ Instructor’s Home Page http://laplace.physics.ubc.ca/People/matt/index.html

Schedule

Tuesday/Thursday, 10:00-11:20 AM -- Hennings 304

Course Links

Syllabus News (last update Sunday September 5, 1:00 PM Course Topics COURSE NOTES Online Course Resources Course Related Software Suggested Hardcopy References Homework Schedule and Problem Sets Suggested Term Projects Student Pages Computer Access Info: [P & A Computer Labs --Schedule Linux Lab (PCs) Hennings 205

COURSE OVERVIEW

This course will provide anintroduction

tutorial syllabus Course Topics

Text: required

notes optional text,Numerical Recipes (2nd edition) et al anticipate doing further numerical work. on-line

the Suggested References Course Resources

Grades: Tests, Homework and Term Projects

may Tests

There will be twoone-hour

Midterm:Thursday, November 2, 10:00 AM TBA Note that the Midterm and Final countequally

Homework Homework will be assigned at least a week before it may Late Homework Policy described below. Homework Schedule

Each homework will contributeroughly Term Projects

a one-page outline of your selected project is due Tuesday, October 19 at the latest You are encouraged to develop your own project ideas, but

now hardcopy of and Astronomy Main Office.

discretion. and as per the Late Work Policy

particular,cannot be MS-Windows specific Late Work Policy (Strictly Enforced)

instructor Submitted homework which no additional action on your part is required

Computer Access

Physics & Astronomy Computer Lab Linux Lab machines vn.physics.ubc.ca

Tutorials

Other Help

how to extract

Syllabus

Tuesday Thursday September 7 September 9

September 14 September 16

September 21 September 23 [H1 due] September 28 September 30

October 5 October 7

October 12 October 14 [H2 due] October 19

[Project outlines due]

October 21

October 26 October 28 [H3 due] November 2 MIDTERM

November 4

November 9 November 11 REMEMBRANCE DAY [NO CLASS] November 16 November 18

November 23 [H4 due]

November 25

November 30 December 2 [HW5 & Term Projects due]

Syllabus Notes

Homework assignments H1 through H5 See Course Topics Term project outlines are dueOCTOBER 19 Term projects are due DECEMBER 2 (last class day)

Other Important Dates

Tuesday, September 21:

Monday, October 11: Friday, October 15:

Thursday, November 11: Remembrance Day. University closed.NO CLASS. Friday, December 3: Tuesday, December 7: Tuesday, December 21:

Calendar Academic Year pages for more information